Author's response to reviews

**Title:** A semiparametric Bayesian proportional hazards model for interval censored data with frailty effects

**Authors:**

- Volkmar Henschel (vhenschel@gmx.de)
- Jutta Engel (engel@ibe.med.uni-muenchen.de)
- Dieter Hoelzel (hoe@ibe.med.uni-muenchen.de)
- Ulrich Mansmann (mansmann@ibe.med.uni-muenchen.de)

**Version:** 2  **Date:** 12 September 2008

**Author's response to reviews:**

I was asked to make changes in the abstract, give information on competing interests, and author's contributions. I also provide the software described in the paper.